SPECIFIC COMMENTS TO AUTHORS
This manuscript studied the cardiovascular risk factors among colon cancer survivors and compared the difference between 2017 and 2007. It suggested that the risk of cardiovascular and cerebrovascular events were increased with years. Also, it studied the prevalence of CVD risk factors. Of all, this manuscript has innovation and guidance for clinical practice.

Reply-
Dear peer reviewer, thank you for reviewing our manuscript and your expert critique.